Author's response to reviews

Title: A self-management program for employees with complaints of the arm, neck or shoulder (CANS): study protocol for a randomized controlled trial.

Authors:

Nathan Hutting (N.Hutting@iq.umcn.nl)
J Bart Staal (B.Staal@iq.umcn.nl)
Yvonne F Heerkens (Yvonne.Heerkens@han.nl)
Josephine A Engels (Josephine.Engels@han.nl)
Maria W G Nijhuis-van der Sanden (R.Nijhuis@iq.umcn.nl)

Version: 6 Date: 30 July 2013

Author's response to reviews: see over
Ref: A self-management program for employees with complaints of the arm, neck or shoulder (CANS): study protocol for a randomized controlled trial, Hutting, Staal, et al.

Dear sirs,

We would like to thank the reviewer for all the constructive comments to improve our manuscript.

The manuscript has been revised according to the reviewers’ comments as follows:
1) our responses/explanations are presented in italics below the reviewers’ comments.
2) we have added new version of our revised manuscript that includes all the amendments

# Editorial board:
1. All the R(eference)s that have translated titles should [(Dutch)] at the end of the R instead of just after the dutch title.
   We changed this.

2. Page 23, R 67. Does not look correct. Should one of the authors be listed instead of the initials [S.C.] and [(1995)] should be deleted as the date is listed twice.
   We changed this.